Physics 5B Introduction to Physics II Winter 2009

Lecture and Exam Schedule for Physics 5B (page 1)

Date Textbook readings HW due Wed Jan 7 13-1 , 13-2 , 13-3 Fri Jan 9 13-4 , 13-5 , 13-6 Mon Jan 12 13-7 , 13-8 , 13-9 Wed Jan 14 13-10 , 13-11 , 13-12 , 13-13 #1 due Fri Jan 16 14-1 , 14-2 Mon Jan 19 MARTIN LUTHER KING JR. DAY (no class) Wed Jan 21 14-3 , 14-4 , 14-5 #2 due Fri Jan 23 14-6 , 14-7 , 14-8 Mon Jan 26 15-1 , 15-2 Wed Jan 28 15-3 , 15-4 , 15-5 #3 due Fri Jan 30 MIDTERM EXAM I (in-class exam) Mon Feb 2 15-6 , 15-7 , 15-8 , 15-9 Wed Feb 4 16-1 , 16-2 , 16-3 #4 due Fri Feb 6 16-4 , 16-5 Mon Feb 9 16-6 , 16-7 Wed Feb 11 16-8 , 32-1 , 32-2 #5 due Fri Feb 13 32-3 , 32-4 Mon Feb 16 PRESIDENTS’ DAY (no class) Wed Feb 18 32-5 , 32-6 , 32-7 , 32-8 #6 due Fri Feb 20 33-1 , 33-2 Mon Feb 23 33-3 , 33-4 , 33-5 Wed Feb 25 33-6 , 33-7 , 33-8 #7 due Fri Feb 27 MIDTERM EXAM II (in-class exam)

Lecture and Exam Schedule for Physics 5B (page 2)

Date Textbook readings HW due Mon Mar 2 34-1 , 34-2 , 34-3 Wed Mar 4 34-4 , 34-5 #8 due Fri Mar 6 35-1 , 35-2 Mon Mar 9 35-3 , 35-4 , 35-5 , 35-6 Wed Mar 11 35-7 , 35-8 , 35-9 #9 due Fri Mar 13 35-10 , 35-11 , 35-13 Mon Mar 16 Course Review #10 due Thu Mar 19 FINAL EXAM

Notes:

1. All textbook readings are from Physics for Scientists & Engineers with Modern Physics, 4th edition, by Douglas C. Giancoli. In the lecture schedule above, 13-1 means Chapter 13, section 1, etc. You are strongly advised to read the assigned sections for a given day prior to the class lecture.

2. If you have purchased the course textbook in two separate volumes, please note that Chapters 13–16 appear in Volume I, and Chapters 32–35 appear in Volume II.

3. All homework (HW) is due on a Wednesday (exception: HW #10 is due on a Monday, the last day of class). Odd numbered homework sets (#1, 3, 5, 7, 9) will cover material from the three lectures prior to the due date. Even numbered homework sets (#2, 4, 6, 8, 10) will cover material from the two lectures prior to the due date.

4. There will be two midterm exams. The first midterm will take place on Friday January 30, and the second midterm will take place on Friday February 27. Both midterm exams will be held in Thimann Lecture Hall 3 during the usual class lecture time. No class lecture will be given on those two days.

5. The final exam will be held on Thursday March 19, 2009 in Thimann Lecture Hall 3 (the location of the course lectures) from 8:00–11:00 am.